Author's response to reviews

Title: Mesh penetrating the cecum and bladder following inguinal hernia surgery: a case report

Authors:
Hiroshi Asano (hiroshia@saitama-med.ac.jp)
Saori Yajima (syajima@saitama-med.ac.jp)
Yoshie Hosoi (yhosoi@saitama-med.ac.jp)
Makoto Takagi (mtakagi@saitama-med.ac.jp)
Hiroyuki Fukano (hirof53530@yahoo.co.jp)
Yasuhiro Ohara (y_ohara@saitama-med.ac.jp)
Nozomi Shinozuka (zukasino@saitama-med.ac.jp)
Takaya Ichimura (taka_ichimu@yahoo.co.jp)

Version: 1 Date: 07 Apr 2017

Author’s response to reviews:

Dear Dr. Rao:

Thank you for reviewing our manuscript entitled “Mesh penetrating the cecum and bladder following inguinal hernia surgery: a case report” which was submitted to the Journal of Medical Case Reports.

We have addressed the issue indicated in the review report.

Our responses to the reviewer’s comment are presented below.

We hope that the revised manuscript will be suitable for publication in your Journal.

-As claimed by the authors, it is not true that their's is the first report of two organ involvement. The authors need to acknowledge this case report and add it to their reference.

We have revised the Background and Discussion, as shown in the manuscript. And added it to our References.